Author’s response to reviews

Title: Utilisation of malaria preventive measures during pregnancy and birth outcomes in Ibadan, Nigeria

Authors:

   Olukemi O Tongo (ktongo2@yahoo.com)
   Adebola E Orimadegun (beorimadegun@yahoo.com)
   Olusegun O Akinyinka (asegun@hotmail.com)

Version: 2 Date: 15 March 2011

Author’s response to reviews:

The title has been ammended as suggested
The abstract corrected and the objective of the study has been included
Key words have been included.
The literature review in the background section has been revised.
Methodology - the recruitment strategy has been explained and regularity of IPTsp explained.
Results - the tables and headings have been adjusted for further clarity, unnecessary statistical tests removed and the necessary multivariate analysis has now been included.
Typographical errors have been corrected